Author’s response to reviews

Title: Situational Awareness within Objective Structured Clinical Examination (OSCE) Stations in Undergraduate Medical Training - A Literature Search

Authors:

Markus Fischer (m.fischer2@nuigalway.ie)

Kieran Kennedy (kieran.kennedy@nuigalway.ie)

Jean Ker (j.s.ker@dundee.ac.uk)

Paul O’Connor (paul.oconnor@nuigalway.ie)

Eva Doherty (edoherty@rcsi.ie)

Thomas Kropmans (thomas.kropmans@nuigalway.ie)

Version: 1 Date: 04 Dec 2016

Author’s response to reviews:

The word introduction was replaced by Background.

Methods section was added to the abstract.